Author’s response to reviews

Title: Limb girdle muscular dystrophy type 2G with myopathic-neurogenic motor unit potentials and a novel muscle image pattern

Authors:

Ana Cotta (cotta@sarah.br)
Julia F Paim (julia@sarah.br)
Antonio L Cunha-Junior (400564@sarah.br)
Rafael X Neto (9679@sarah.br)
Simone V Nunes (7713@sarah.br)
Monica M Navarro (400039@sarah.br)
Jaquelin Valicek (400585@sarah.br)
Elmano Carvalho (9230@sarah.br)
Lydia U Yamamoto (lydiauy@usp.br)
Camila F Almeida (camila.almeida@usp.br)
Shelida V Braz (12758@sarah.br)
Reinaldo I Takata (202911@sarah.br)
Mariz Vainzof (mvainzof@usp.br)

Version: 2
Date: 22 July 2014

Author’s response to reviews:

Dear Celine Zapanta,

Journal Editorial Office - BioMed Central

As requested, the article sections have been adequated to the journal’s instructions in the website as:

1) Title page (two pages with a list with the full names, institutional addresses and email addresses for all authors and the corresponding author), Abstract, Keywords, Background, Case presentation, Conclusions, Consent, List of abbreviations, Competing interests, Author’s contributions, Author’s informations, Acknowledgments, Endnotes (none), References, Illustrations and figures, Tables and captions (none), Additional files.

2) A list of abbreviations was included in this new version of the manuscript.

3) According to the instructions, the Case presentation section "may be broken into subsections with appropriate subheadings", therefore, the Case presentation section was divided in the following subheadings: Patient Information, Clinical Findings, Diagnostic Assessment, and Discussion.

Yours faithfully,

Mariz Vainzof